Author’s response to reviews

Title: Knowledge and risk behaviors related to HIV/AIDS, and their association with information resource among Men who have sex with men in Heilongjiang province, China

Authors:

Shengyuan Liu (liushenglb@126.com)
Kaili Wang (kaili_w@126.com)
Songpo Yao (guangenzhou@163.com)
Xiaotong Guo (guo.xiaotong@163.com)
Yancheng Liu (yancheng_liu@yahoo.cn)
Binyou Wang (chivalry0683@hotmail.com)

Version: 7 Date: 6 May 2010

Author’s response to reviews: see over
Response to the Editor

Dear Editor:

Thank you very much for reviewing our manuscript entitled “Knowledge and risk behaviors related to HIV/AIDS, and their association with information resource among Men who have sex with men in Heilongjiang province, China.”

We are sincerely grateful to the Editor and the reviewers for our critical and helpful corrections, as well as helpful comments. We revised our manuscript according to the reviewer’s suggestions and comments as described in point-by-point reply and marked them in red on the responding letter.

1. Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is “First name Initial Last name” (e.g. Joe F Bloggs)

   Answer: we have changed the name in style and order.

2. Table 3: tables should fit into an A4 page (maximum size 176mm x 230mm) and appear in portrait format. If you cannot re-orientate the table to fit this size, please move it to the additional files. Tables in the additional files will be made available with the final manuscript upon publication but will not be included in the manuscript in situ. Please ensure that you update the text to cite additional files and include a legend in the additional files section.

   Answer: we have moved the table 3 to the additional files, updated the text to cite additional files and included a legend in the additional files section.